Author’s response to reviews

Title: Assessment of a new questionnaire for self-reported sun sensitivity in an occupational skin cancer screening program

Authors:

Jürgen Tacke (juergentacke@yahoo.de)
Jens Dietrich (dietrichjens@gmx.de)
Beatrix Steinebrunner (b.steinebrunner@yahoo.de)
Antonius Reifferscheid (antonius.reifferscheid@henkel.com)

Version: 6 Date: 25 September 2008

Author's response to reviews: see over
RE: Fourth revision
MS: 1413874163189489
TITLE: Assessment of a new questionnaire for self-reported sun sensitivity in an occupational skin cancer screening program
AUTHORS: Jürgen Tacke, Jens Dietrich, Beatrix Steinebrunner and Antonius Reifferscheid

September 25, 2008

Dear BMC Editors,

We have found transposed digits on page 6 and made the necessary changes. The manuscript is now ready for publication.

Thank you again for accepting our manuscript for publication in BMC Dermatology.

Sincerely,

Jürgen Tacke, M.D., M.P.H.(USA)
Dermatologist
Joint Group Practice Dermatology
Bremsstrasse 19
50969 Cologne
Email: juergen.tacke@googlemail.com
Phone: +49-221-936414-0
Fax: +49-221-936414-3